%\def\paperauthorB{Author_B_fullname}
% you can add more authors using alphabets after \paperauthor
%------------------------------------------------------------------------------------------
% Do not delete this part starting from here and up to .....
%\setcounter{page}{1}
%\input{../../expages.tex}\setpagenumber{1}
%\input{expages.tex}\setpagenumber{1}
% Saves a lot of ouptut space in PDF... after conversion with the distiller	
% Delete if you cannot get PS fonts working on your system.
% pdf-tex settings: detect automatically if run by latex or pdflatex
% compiling with pdflatex
% compiling with latex
% hyperrefs are active in the pdf file after conversion
%------------------------------------------------------------------------------------------
% This command replaces \name{The DAFx Crew}
%\usepackage{layout}
%------------------------------------------------------------------------------------------
%%%%%%%%%%%%%%%%%%%%%%% to here %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\documentclass[oneside]{article}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{icma2021,amssymb,amsmath,mathrsfs}
\usepackage{mathptmx}
\usepackage[pdftex,
pdftitle={\papertitle},
colorlinks=false, bookmarksnumbered, pdfstartview= XYZ	]{hyperref}
\usepackage[pdftex]{graphicx}
\usepackage[figure,table]{hypcap}
\usepackage[dvips]{graphicx}
\usepackage[dvips,
colorlinks=false,           bookmarksnumbered, pdfstartview= XYZ	]{hyperref}
\usepackage[figure,table]{hypcap}

\setcounter{MaxMatrixCols}{10}
%TCIDATA{OutputFilter=Latex.dll}
%TCIDATA{Version=5.50.0.2953}
%TCIDATA{<META NAME="SaveForMode" CONTENT="1">}
%TCIDATA{BibliographyScheme=Manual}
%TCIDATA{LastRevised=Wednesday, November 24, 2021 17:29:18}
%TCIDATA{<META NAME="GraphicsSave" CONTENT="32">}

\def\papertitle{Full title of the paper}
\def\paperauthorA{AuthorAfullname}
\ninept
\newif\ifpdf
\ifx\pdfoutput\relax
\else
\ifcase\pdfoutput
\pdffalse
\else
\pdftrue
\fi
\ifpdf
\pdfcompresslevel=9
\else
\fi
\newcommand\dsum{\displaystyle\sum}
\newcommand\dprod{\displaystyle\prod}
\affiliation{Mohamed Houasni$^{1}$ and Salah Zitouni$^{2}$}
{$^{1}$ Facult\'{e} des Sciences et de la Technologie, Universit\'{e} DBKM,\\
Khemis miliana, Alg\'{e}rie\\
$^{2}$ Department of Mathematics and Informatics  \\
Souk Ahras Univ, P.O. Box 1553,
Souk Ahras, 41000, Algeria}



\begin{document}
\begin{thebibliography}{99}
\bibitem{11} M.S. Alves, C. Buriol , M.V. Ferreira et al., Asymptotic
behaviour for the vibrations modeled by the standard linear solid solid
modelwith a thermal effect, \emph{J Math Anal Appl} 399, 472--479, (2013).

\bibitem{1} M.S. Alves, P. Gamboa, G. C. Gorain et al., Asymptotic behavior
of a flexible structure with Cattaneo type of thermal effect. \emph{Indag
Math}., 27(3), 821--834, (2016).

\bibitem{ap} T. A. Apalara, On the stability of porous-elastic system with
microtemparatures, \emph{Journal of Thermal Stresses}, 42$\left( 2\right) $,
265--278, $\left( 2019\right) .$

\bibitem{bre} H. Brezis, Functional Analysis, Sobolev Spaces and Partial
Differential Equations. \emph{Springer Science + Business Media, LLC},
(2011). doi 10.1007/978-0-387-70914-7

\bibitem{cas} P. S. Casas, R. Quintanilla, Exponential stability in
thermoelasticity with microtemperatures, \emph{International Journal of
Engineering Science}, 43, 33--47, (2005).

\bibitem{ch2} S. Chirit$\widetilde{a}$, M. Ciarletta, C. D'Apice, On the
theory of thermoelasticity with microtemperatures, \emph{J. Math. Anal.
Appl.,} 397, 349--361, (2013)

\bibitem{ch} S. Chirit$\widetilde{a}$, A. Danescuca, On the propagation
waves in the theory of thermoelasticity withmicrotemperatures, \emph{%
Mechanics Research Communications,} 75, 1--12, (2016).

\bibitem{dd1} R. Datko, J. Lagnese, M. P. Polis, An example on the effect of
time delays in boundary feedback stabilization of wave equations, \emph{SIAM
J. Control Optim.} , 24, 152-156, $\left( 1986\right) $.

\bibitem{dri} H. Dridi, A. Djebabla, On the stabilization of linear porous
elastic materials by microtemperature effect and porous damping, \emph{%
ANNALI DELL'UNIVERSITA' DI FERRARA, }(2019).
https://doi.org/10.1007/s11565-019-00333-2

\bibitem{2g} G. C. Gorain, Exponential stabilization of longitudinal
vibrations of an inhomogeneous beam, \emph{Nonlinear Oscil}., 16, 157--164,
(2013).

\bibitem{grot} R. Grot, Thermodynamics of a continuum with microstructure,
Int. J. Eng. Sci., 7, 801--814, (1969).

\bibitem{hach} M. Hachelfi, A. Djebabla, N. Tatar, On the decay of the
energy for linear thermoelastic systems by thermal and micro-temperature
effects, \emph{Eurasian Journal of Mathematical and Computer Applications}, 6%
$\left( 4\right) $, 29--37, (2018).

\bibitem{isan6} D. Ie\c{s}an, Thermoelasticity of bodies with microstructure
and microtemperatures, \emph{Internat. J. Solids Structures,} 44,
8648--8662, (2007).

\bibitem{isan3} D. Ie\c{s}an, On a theory of thermoelasticity without energy
dissipation for solids with microtemperatures, \emph{Z Angew Math Mech.},
1--16, (2018).

\bibitem{iq} D. Ie\c{s}an, R. Quintanilla, On the Theory of Thermoelasticity
with Microtemperatures, \emph{J. Thermal Stresses,} 23, 199--215, (2000).

\bibitem{isan4} D. Ie\c{s}an, R. Quintanilla, On thermoelastic bodies with
inner structure and microtemperatures, \emph{J. Math. Anal. Appl.} 354,
12--23, (2009).

\bibitem{isan2} D. Ie\c{s}an, R. Quintanilla, Qualitative properties in
strain gradient thermoelasticity with microtemperatures, \emph{Math. Mech.
Solids,} 23, 240--258, (2018).

\bibitem{lii} G. Li, Y. Luan, J. Yu et al., Well-posedness and exponential
stability of a flexible structure with second sound and time delay, \emph{%
APPLICABLE ANALYSIS, }98$\left( 16\right) $, 2903-2915, $\left( 2019\right)
. $

\bibitem{mag} A. Mag$\widetilde{a}$na, R. Quintanilla, Exponential stability
in type III thermoelasticity with microtemperatures, \emph{Z. Angew. Math.
Phys.,} 69$\left( 129\right) $, (2018).

\bibitem{3m} S. Misra , M.S. Alves, G. Gorain et al., Stability of the
vibrations of an inhomogeneous flexible structure with thermal effect, \emph{%
\ Int. J. Dyn. Control,} 3(4), 354--362, (2015).

\bibitem{Miil} D. S. Mitrinovic, J. E. Pecaric, A. M. Fink, Inequalities
involving functions and their integrals and derivatives. \emph{Dordrecht:}
\emph{Kluwer Academic Publishers}, \emph{Springer-Science + Business Media, }
53, (1991).\emph{\ }ISNB 978-94-010-5578-9.

\bibitem{2444} S. Nicaise, C. Pignotti, Stabilization of the wave equation
with boundary or internal distributed delay, \emph{Diff. Int. Equs., }, 21
(9--10), 935--958, $\left( 2008\right) $.

\bibitem{Ri} R. Racke, Instability of coupled systems with delay, \emph{%
Comm. Pure, Appl. Anal.}, 11 (5), 1753-1773, $\left( 2012\right) $.

\bibitem{riha} P. Riha, On the theory of heat-conducting micropolar fluids
with microtemperatures, \emph{Acta Mechanica}, 23$\left( 1\right) ,$ $\left(
1975\right) $.

\bibitem{riha2} P. Riha, On the microcontinuum model of heat conduction in
materials with inner structure, \emph{Int. J. Eng. Sci.}, 14 $\left(
529\right) $, $\left( 1976\right) $.

\bibitem{saci} M. Saci, A. Djebabla, On the stability of linear porous
elastic materials with microtemperatures effects,\emph{\ Journal of Thermal
Stresses}, (2019). DOI: 10.1080/01495739.2020.1779629

\bibitem{sca} A. Scalia, M. Svanadze, On the Representations of Solutions of
the Theory of Thermoelasticity with Microtemperatures, \emph{Journal of
Thermal Stresses}, 29, 849--863, (2006).

\bibitem{sva} M. Svanadze, Fundamental Solutions of the Equations of the
Theory of Thermoelasticity with Microtemperatures, \emph{J. Thermal Stresses}%
, 27, 151--170, (2004).
\end{thebibliography}


\title{Exponential stability for a delayed flexible structure }
\author{}
\maketitle

\begin{abstract}
In this paper, we consider a non-uniform and delayed flexible structure with
temperature and micro-temperature effects. We prove the well-posed of the
problem using semi-groups theory, as well as an exponential stability using
the multiplier method under a small condition on time delay.
\end{abstract}

%\layout

% more pdf-tex settings:
\ifpdf % used graphic file format for pdflatex
\DeclareGraphicsExtensions{.png,.jpg,.pdf}  \else
% used graphic file format for latex
\DeclareGraphicsExtensions{.eps}  \fi

\section{Introduction}

In this paper, we aim to study the following inhomogeneous flexible
structure system with micro-temperature effect:
\begin{equation}
\left\{
\begin{array}{l}
m(x)u_{tt}-(p(x)u_{x}+2\delta (x)u_{xt})_{x}+dw_{x}+\eta \theta _{x}+\mu
u_{t}\left( x,t-\tau _{0}\right) =0, \\
c\theta _{t}-k\theta _{xx}+\eta u_{tx}+k_{1}w_{x}=0, \\
\tau w_{t}-k_{3}w_{xx}+k_{2}w+k_{1}\theta _{x}+du_{tx}=0,%
\end{array}%
\right.  \label{11m.}
\end{equation}%
where $u(x,t)$ is the displacement of a particle at position $x\in (0,L)$
and time $t>0,$ $\theta $ and $w$ are the temperature of the body and the
micro-temperature vector respectively. $\eta >0$ is the coupling constant,
that accounts for the heating effect, and $k,k_{1},k_{2},k_{3},c,d,\tau >0.$
$m(x),$ $\delta (x)$ and $p(x)$ are responsible for the non-uniform
structure of the body, and, respectively, denote mass per unit length of
structure, coefficient of internal material damping and a positive function
related to the stress acting on the body at a point $x$. We consider the
following initial and boundary conditions:%
\begin{equation}
\begin{array}{l}
u\left( .,0\right) =u_{0}\left( x\right) ,\text{ }u_{t}\left( .,0\right)
=u_{1}\left( x\right) ,\text{ }\theta \left( .,0\right) =\theta _{0}\left(
x\right) ,w\left( .,0\right) =w_{0}\left( x\right) ,\text{ \ }\forall x\in
\lbrack 0,L] \\
u\left( 0,t\right) =u\left( L,t\right) =\theta \left( 0,t\right) =\theta
\left( L,t\right) =w_{x}\left( 0,t\right) =w_{x}\left( L,t\right) =0,\text{ }%
\forall t\geq 0.%
\end{array}
\label{12m.}
\end{equation}

In the presence of second sound, Alves et al. \cite{1} concerned with the
system%
\begin{equation}
\left\{
\begin{array}{l}
m(x)u_{tt}-(p(x)u_{x}+2\delta (x)u_{xt})_{x}+\eta \theta _{x}=0, \\
\theta _{t}+kq_{x}+\eta u_{tx}=0, \\
\tau q_{t}+\beta q+k\theta _{x}=0,%
\end{array}%
\right.  \label{1334.}
\end{equation}%
They established the well-posedness of the system and proved its stability
exponential and polynomial under suitable boundary conditions. We know that
the dynamic systems with delay terms have become a major research subject in
differential equation since the $1970s$ of the last century. The delay
effect that is similar to memory processes is important in the research of
applied mathematics such as physics, non-instant transmission phenomena and
biological motivations. Moreover, Datko et al. \cite{dd1} in 1986 showed
that the presence of the delay may not only destabilize a system which is
asymptotically stable in the absence of the delay but may also lead to
ill-posedness (see also \cite{2444} and \cite{Ri}). Li et al. \cite{lii}
considered (\ref{1334.}) with a delay term of the form $\mu u_{t}\left(
x,t-\tau _{0}\right) $ in its first equation,
\begin{equation}
\left\{
\begin{array}{l}
m(x)u_{tt}-(p(x)u_{x}+2\delta (x)u_{xt})_{x}+\eta \theta _{x}+\mu
u_{t}\left( x,t-\tau _{0}\right) =0, \\
\theta _{t}+kq_{x}+\eta u_{tx}=0, \\
\tau q_{t}+\beta q+k\theta _{x}=0,%
\end{array}%
\right.
\end{equation}%
they proved that the system is exponential decay under a "small" condition
on time delay. For more details discussion on the subject see \cite{11,2g}
and the references therein.

The linear theory of thermo-elasticity with micro-temperatures for materials
with inner structure whose particles, in addition to the classical
displacement and temperature fields, possess micro-temperatures was
constructed by Ie\c{s}an and Quintanilla \cite{iq,isan2}. The work is
motivated by increasing use of materials which possess thermal variation at
a microstructure level. The same authors proved an existence theorem and
established the continuous dependence of solutions of the initial data and
body loads. Originally, Grot \cite{grot} was the first to take into
consideration the inner structure of a body in order to develop a
thermodynamic theory for thermo-elastic materials where micro-elements, in
addition to classic micro-deformations, possess micro-temperatures. While,
the fundamental solution of the equations of the theory of thermo-elasticity
with micro-temperatures was constructed by Svanadze \cite{sva}. Riha \cite%
{riha,riha2} developed a further study concerning heat conduction in
thermo-elastic materials with inner structure. We note that the concept of
micro-temperature was just used in the theory of thermodynamics for elastic
materials with microstructure. In addition to micro-deformations of the
string, the micro-elements of the continuum possess micro-temperatures which
represent the variation of the temperature within a micro-volume. We refer
the interested readers to \cite%
{ap,cas,ch2,ch,dri,hach,isan6,isan3,isan4,mag,saci,sca} for details
discussion on the theory.

Motivated by works mentioned above, we investigate (\ref{11m.})-(\ref{12m.})
under suitable condition and establish the well-posedness of the problem
using semi-group theory, as well as the stability result of the solution
using the multiplier method. Our purpose in the present manuscript is to
obtain an exponential decay rate estimates of the energy function of (\ref%
{11m.}) with a "small" condition on time delay. Introducing the
microtemperature damping instead of the second sound together with the delay
term in the internal feedback of flexible structure makes our problem
different from those considered so far in the literature.

This paper is organized as follows; In the second section, we introduce some
assumptions needed in our work then prove the well-posedness of the system (%
\ref{11m.})-(\ref{12m.}). In the last section we state and prove our
stability result.

\section{Well-posedness of the problem}

In this section, we present some assumptions and give the existence and
uniqueness result of system (\ref{11m.})-(\ref{12m.}) using the semigroup
theory. Throughout this paper, $c^{\prime }$ represents a generic positive
constant and is different in various occurrences.

Taking the following new variable
\begin{equation*}
z(x,\rho ,t)=u_{t}\left( x,t-\rho \tau _{0}\right) ,\text{ in }\left(
0,L\right) \times \left( 0,1\right) \times \left( 0,\infty \right) ,
\end{equation*}%
then we obtain%
\begin{equation*}
\left\{
\begin{array}{l}
sz_{t}(x,\rho ,t)+z_{\rho }(x,\rho ,t)=0, \\
z(x,0,t)=u_{t}(x,t).%
\end{array}%
\right.
\end{equation*}%
Consequently, problem (\ref{11m.})-(\ref{12m.}) is equivalent to%
\begin{equation}
\left\{
\begin{array}{l}
\begin{array}{l}
m(x)u_{tt}-(p(x)u_{x}+2\delta (x)u_{xt})_{x}+dw_{x}+\eta \theta _{x}+\mu
z(x,1,t)=0, \\
c\theta _{t}-k\theta _{xx}+\eta u_{tx}+k_{1}w_{x}=0, \\
\tau w_{t}-k_{3}w_{xx}+k_{2}w+k_{1}\theta _{x}+du_{tx}=0, \\
sz_{t}(x,\rho ,s,t)+z_{\rho }(x,\rho ,s,t)=0,%
\end{array}
\\
\begin{array}{l}
u\left( .,0\right) =u_{0}\left( x\right) ,\text{ }u_{t}\left( .,0\right)
=u_{1}\left( x\right) ,\text{ }\theta \left( .,0\right) =\theta _{0}\left(
x\right) ,w\left( .,0\right) =w_{0}\left( x\right) , \\
u\left( 0,t\right) =u\left( L,t\right) =\theta \left( 0,t\right) =\theta
\left( L,t\right) =w_{x}\left( 0,t\right) =w_{x}\left( L,t\right) =0,\text{ }%
\forall t\geq 0, \\
z(x,0,t)=u_{t}(x,t) \\
z(x,\rho ,0)=f_{0}(x,-\rho \tau _{0})%
\end{array}%
\end{array}%
\right.  \label{11m..}
\end{equation}%
where $f_{0}$ is the history function.

The aim of this section is to prove that system (\ref{11m.}) is well-posed.
From Equation (\ref{11m.})$_{3}$ and the boundary conditions, we have%
\begin{equation*}
\frac{d}{dt}\int_{0}^{L}w\left( x,t\right) dx+\frac{k_{2}}{\tau }%
\int_{0}^{L}w\left( x,t\right) dx=0,\text{ }\forall t\geq 0,
\end{equation*}%
thus%
\begin{equation*}
\int_{0}^{L}w\left( x,t\right) dx=\left( \int_{0}^{L}w_{0}dx\right) \exp
\left( \frac{-t}{\tau }k_{2}\right) ,\text{ }\forall t\geq 0,
\end{equation*}%
So, if we set%
\begin{equation*}
\widetilde{w}\left( x,t\right) =w\left( x,t\right) -\frac{1}{L}\left(
\int_{0}^{L}w_{0}dx\right) \exp \left( \frac{-t}{\tau }k_{2}\right),
\end{equation*}
\begin{equation*}
\text{ } t\geq 0,\text{ }x\in \lbrack 0,L],
\end{equation*}
then, $(u,u_{t},\theta ,\widetilde{w},z)$ satisfies Equation (\ref{11m.}),
and
\begin{equation*}
\int_{0}^{L}\widetilde{w}\left( x,t\right) dx=0,
\end{equation*}%
for all $t\geq 0$. In the sequel, we shall work with $\widetilde{w}$ but we
write $w$ for simplicity.

The energy functional associated to (\ref{11m..}), namely,%
\begin{equation}
\mathcal{E}(t)=\frac{1}{2}\int_{0}^{L}\left\{
p(x)u_{x}^{2}+m(x)u_{t}^{2}+c\theta ^{2}+\tau w^{2}\right\} dx+\frac{%
\left\vert \mu \right\vert \tau _{0}}{2}\int_{0}^{L}\int_{0}^{1}z^{2}(x,\rho
,t)d\rho dx,  \label{ett.}
\end{equation}%
we denote $\mathcal{E}(t)=\mathcal{E}\left( t,u,u_{t},\theta ,w,z\right) $
and $\mathcal{E}(0)=\mathcal{E}\left( 0,u_{0},u_{1},\theta
_{0},w_{0},f_{0}\right) $ for simplicity of notations. Then the energy $E$
is decreasing function and satisfies, for and all $t\geq 0,$%
\begin{eqnarray*}
\mathcal{E}^{\prime }(t) &=&-2\int_{0}^{L}\delta \left( x\right)
u_{xt}^{2}dx-k_{2}\int_{0}^{L}w^{2}dx-k_{3}\int_{0}^{L}w_{x}^{2}dx-k%
\int_{0}^{L}\theta _{x}^{2}dx \\
&&-\mu \int_{0}^{L}u_{t}z_{1}(x,1,t)dx \\
&\leq &\left( \left\vert \mu \right\vert -2\delta (\xi _{6})/l\right)
\int_{0}^{L}u_{t}^{2}dx-k_{2}\int_{0}^{L}w^{2}dx-k_{3}%
\int_{0}^{L}w_{x}^{2}dx-k\int_{0}^{L}\theta _{x}^{2}dx.
\end{eqnarray*}%
To obtain precise decay rates of $E(t)$ as $t\rightarrow +\infty ,$ we
assume that

\begin{equation}
m,\text{ }\delta ,\text{ }p\in W^{1,\infty }(0,L),\text{ }m(x),\text{ }p(x)%
\text{, }\delta (x)>0,\text{ \ }\forall x\in \lbrack 0,L].  \label{h2.}
\end{equation}

Let us introducing the vector function $U=(u,v,\theta ,w,z)^{T}$, where $%
v=u_{t}$, using the standard Lebesgue space $L^{2}(0,L)$ and the Sobolev
space $H_{0}^{1}(0,L)$ with their usual scalar products and norms for define
the spaces:%
\begin{equation*}
\mathcal{H}:=H_{0}^{1}(0,L)\times \left[ L^{2}(0,L)\right] ^{2}\times
L_{\ast }^{2}(0,L)\times L^{2}((0,L)\times (0,1)),
\end{equation*}%
and%
\begin{equation*}
H_{\ast }^{2}(0,L)=\left\{ w\in H^{2}(0,L):w_{x}\left( L\right) =w_{x}\left(
0\right) =0\right\} ,
\end{equation*}%
where%
\begin{equation*}
L_{\ast }^{2}(0,L)=\left\{ w\in L^{2}(0,L):\int_{0}^{L}w\left( s\right)
ds=0\right\} .
\end{equation*}%
We equip $\mathcal{H}$ with the inner product%
\begin{eqnarray*}
(U,\widetilde{U})_{\mathcal{H}} &=&\int_{0}^{L}p\left( x\right) u_{x}%
\widetilde{u}_{x}dx+\int_{0}^{L}m\left( x\right) v\widetilde{v}%
dx+c\int_{0}^{L}\theta \widetilde{\theta }dx+\tau \int_{0}^{L}w\widetilde{w}%
dx \\
&&+\tau _{0}\left\vert \mu \right\vert \int_{0}^{L}\int_{0}^{1}z\widetilde{z}%
(x,\rho ,t)d\rho dx.
\end{eqnarray*}

Next, the system (\ref{11m..}) can be reduced to the following abstract
Cauchy problem:%
\begin{equation}
\left\{
\begin{array}{ll}
U^{\prime }(t)+\left( \mathcal{A+B}\right) U(t)=0, & t>0 \\
U(0)=U_{0}=(u_{0},u_{1},\theta _{0},w_{0},f_{0})^{T}, &
\end{array}%
\right.  \label{proo.}
\end{equation}%
where the operator $\mathcal{A}:D(\mathcal{A})\rightarrow \mathcal{H}$ is
defined by%
\begin{equation*}
\mathcal{A}U=\left(
\begin{array}{c}
-v \\
-\frac{1}{m(x)}\left[ (p(x)u_{x}+2\delta (x)v_{x}-\eta \theta -dw)_{x}-\mu
z(x,1,t)-\left\vert \mu \right\vert v\right] \\
\frac{1}{c}\left( -k\theta _{xx}+\eta u_{tx}+k_{1}w_{x}\right) \\
\frac{1}{\tau }(-k_{3}w_{xx}+k_{2}w+k_{1}\theta _{x}+du_{tx})%
\end{array}%
\right) ,
\end{equation*}%
\begin{equation*}
\mathcal{B}U=\left(
\begin{array}{c}
0 \\
-\frac{\left\vert \mu \right\vert v}{m(x)} \\
0 \\
0%
\end{array}%
\right) ,
\end{equation*}%
The domain of $\mathcal{A}$ is then%
\begin{equation*}
D(\mathcal{A})=\left\{
\begin{array}{l}
U\in \mathcal{H}\mid u\in H^{2}(0,L)\cap H_{0}^{1}(0,L),v\in H_{0}^{1}(0,L),%
\text{ }\theta \in H^{2}(0,L), \\
w\in L_{\ast }^{2}(0,L)\cap H_{\ast }^{2}(0,L),\text{ }z,z_{\rho }\in
L^{2}((0,L)\times (0,1)),z(x,0)=u_{t}(x)%
\end{array}%
\text{ }\right\} ,
\end{equation*}%
which is dense in $\mathcal{H}$.

\begin{proposition}
\label{Propo} Let $U_{0}\in \mathcal{H}$ be given. Problem (\ref{proo.})
possesses then a unique solution satisfying $U\in C\left(
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
^{+};\mathcal{H}\right) .$ If $U_{0}\in D(\mathcal{A}),$ then $U\in
C^{1}\left(
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
^{+};\mathcal{H}\right) \cap C\left(
%TCIMACRO{\U{211d} }%
%BeginExpansion
\mathbb{R}
%EndExpansion
^{+};D(\mathcal{A})\right) .$
\end{proposition}

\begin{proof}
Using semi-group theory. We prove that $\mathcal{A}$ is a maximal monotone
operator and $\mathcal{B}$ is Lipschitz continuous in $\mathcal{H}.$
Consequently, $\mathcal{A+B}$ is the infinitesimal generator of a linear
contraction C$_{0}$-semigroup on $\mathcal{H}$.
\end{proof}

\section{Exponential stability}

In this section, we introduce some lemmas allow us to achieve our goal,
which is the proof of the stability result.

\begin{lemma}
\cite{Miil} (Poincar\'{e} type Scheeffer's inequality): Let $h\in $ $%
H_{0}^{1}(0,L)$. Then it holds%
\begin{equation}
\int_{0}^{L}\left\vert h\right\vert ^{2}dx\leq \text{ }l\int_{0}^{L}\left\vert h_{x}\right\vert ^{2}dx,\text{ \ }l=\frac{L^{2}}{\pi ^{2}}.
\label{Muu.}
\end{equation}
\end{lemma}

\begin{lemma}
\cite{1,3m} \label{mean.}: Let $(u,u_{t},\theta ,w)$ be the solution to
system (\ref{11m.})-(\ref{12m.}), with an initial datum in $D(\mathcal{A})$.
Then, for any $t>0$, there exists a sequence of real numbers (depending on
t), denoted by $\xi _{i}\in \lbrack 0,L](i=1,...,6)$, such that:%
\begin{equation*}
\int_{0}^{L}p\left( x\right) u_{x}^{2}dx=p(\xi _{1})\int_{0}^{L}u_{x}^{2}dx,%
\text{ }\int_{0}^{L}\delta \left( x\right) u^{2}dx=\delta (\xi
_{4})\int_{0}^{L}u^{2}dx,
\end{equation*}%
\begin{equation*}
\text{ \ }\int_{0}^{L}m\left( x\right) u_{t}^{2}dx=m(\xi
_{2})\int_{0}^{L}u_{t}^{2}dx,\text{ }\int_{0}^{L}\delta \left( x\right)
u_{x}^{2}dx=\delta (\xi _{5})\int_{0}^{L}u_{x}^{2}dx,
\end{equation*}%
\begin{equation*}
\int_{0}^{L}m\left( x\right) u^{2}dx=m(\xi _{3})\int_{0}^{L}u^{2}dx,\text{ }%
\int_{0}^{L}\delta \left( x\right) u_{xt}^{2}dx=\delta (\xi
_{6})\int_{0}^{L}u_{xt}^{2}dx.
\end{equation*}
\end{lemma}

\begin{lemma}
Let $\left( u,v,\theta ,w,z\right) $ be the solution of (\ref{11m..}), then
the energy $\mathcal{E}$ is non-increasing function and satisfies, for all $%
t\geq 0,$%
\begin{eqnarray}
\mathcal{E}^{\prime }(t) &=&-2\int_{0}^{L}\delta \left( x\right)
u_{xt}^{2}dx-k_{2}\int_{0}^{L}w^{2}dx-k_{3}\int_{0}^{L}w_{x}^{2}dx-k%
\int_{0}^{L}\theta _{x}^{2}dx-\mu \int_{0}^{L}u_{t}z_{1}(x,1,t)dx  \notag \\
&\leq &\left( \left\vert \mu \right\vert -2\delta (\xi _{6})/l\right)
\int_{0}^{L}u_{t}^{2}dx-k_{2}\int_{0}^{L}w^{2}dx-k_{3}%
\int_{0}^{L}w_{x}^{2}dx-k\int_{0}^{L}\theta _{x}^{2}dx.  \label{ET''}
\end{eqnarray}
\end{lemma}

\begin{proof}
Multiplying the equations in (\ref{11m..})$_{1}$,(\ref{11m..})$_{2}$, (\ref%
{11m..})$_{3}$ and (\ref{11m..})$_{4}$ by $u_{t}$, $\theta ,$ $w$ and $%
\left\vert \mu \right\vert z$, respectively, integrate over $(0,L)$ and $%
(0,L)\times (0,1)$ and using (\ref{Muu.}), we obtain (\ref{ET''}).
\end{proof}

\begin{lemma}
The functional%
\begin{equation}
I_{1}\left( t\right) :=\int_{0}^{L}\left( \delta
(x)u_{x}^{2}+m(x)u_{t}u\right) dx,  \label{I1m.}
\end{equation}%
satisfies%
\begin{eqnarray}
I_{1}^{\prime }\left( t\right)  &\leq &-\left( p(\xi _{1})-\mu ^{2}l-\left(
\eta +d\right) \epsilon _{1}\right) \int_{0}^{L}u_{x}^{2}dx+m(\xi
_{2})\int_{0}^{L}u_{t}^{2}dx+\frac{\eta }{4\epsilon _{1}}\int_{0}^{L}\theta
^{2}dx  \notag \\
&&+\frac{d}{4\epsilon _{1}}\int_{0}^{L}w^{2}dx+\frac{1}{4}%
\int_{0}^{L}z^{2}(x,1,t)dx,  \label{I1'}
\end{eqnarray}%
for any $\varepsilon _{1}>0.$
\end{lemma}

\begin{proof}
Differentiating Equation (\ref{I1m.}) with respect to $t$ and using
Equations (\ref{11m.})$_{1}$. By using Young's inequality (see \cite{bre} p.
92) and Lemma \ref{mean.}, we complete the proof.
\end{proof}

\begin{lemma}
The functional%
\begin{equation}
I_{2}(t):=\tau c\int_{0}^{L}\theta \left( \int_{0}^{x}w(y)dy\right) dx,
\label{f44.}
\end{equation}%
satisfies%
\begin{eqnarray}
I_{2}^{\prime }\left( t\right)  &\leq &\left( -k_{1}c+3\varepsilon
_{2}\right) \int_{0}^{L}\theta ^{2}dx+\frac{1}{2\varepsilon _{2}}%
\int_{0}^{L}u_{t}^{2}dx+\frac{1}{4\varepsilon _{2}}\int_{0}^{L}\theta
_{x}^{2}dx  \notag \\
&&+\left( k_{1}\tau +2\varepsilon _{2}c^{\prime }+c^{\prime }\right)
\int_{0}^{L}w^{2}dx+\frac{1}{4\varepsilon _{2}}\int_{0}^{L}w_{x}^{2}dx,
\label{f4m.}
\end{eqnarray}%
for any $\varepsilon _{2}>0.$
\end{lemma}

\begin{proof}
Taking the derivative of (\ref{f44.}) and using (\ref{11m.})$_{2}$ and (\ref%
{11m.})$_{3}$, Integration by parts and using Young's inequality, we infer (%
\ref{f4m.}).
\end{proof}

\begin{lemma}
The functional%
\begin{equation}
I_{3}(t)=\tau _{0}\int_{0}^{L}\int_{0}^{1}e^{-2\tau _{0}\rho }z^{2}(x,\rho
,t)d\rho dx  \label{r}
\end{equation}

satisfies%
\begin{equation}
I_{3}^{^{\prime }}(t)\leq -2I_{3}(t)-c^{\prime
}\int_{0}^{L}z_{1}^{2}(x,1,t)dx+\int_{0}^{L}\varphi _{t}^{2}dx  \label{dr}
\end{equation}

for $c^{\prime }>0.$

\begin{proof}
By differentiating $I_{3}$, then using (\ref{11m..})$_{5}$ and integrating
by parts.
\end{proof}
\end{lemma}

Next, we define a Lyapunov functional $\mathcal{L}$ and show that it is
equivalent to the energy functional.

\begin{lemma}
For $N$ sufficiently large, the functional defined by%
\begin{equation}
\mathcal{L}(t):=N\mathcal{E}\left( t\right)
+I_{1}(t)+N_{1}I_{2}(t)+N_{2}I_{3}(t).  \label{F(t)m.}
\end{equation}%
where $N$ and $N_{1}$ are positive real numbers to be chosen appropriately
later, satisfies%
\begin{equation}
c_{1}^{\prime }\mathcal{E}(t)\leq \mathcal{L}(t)\leq c_{2}^{\prime }\mathcal{%
E}(t),\text{ \ \ }\forall t\geq 0.  \label{equm.}
\end{equation}%
where $c_{1}^{\prime }$ and $c_{2}^{\prime }$ are positive constants.
\end{lemma}

Now, we are ready to state and prove the main result of this section.

\begin{theorem}
Let $\left( u,v,\theta ,w,z\right) $ be the solution of (\ref{11m..}), then
the energy $\mathcal{E}$ satisfies, for all $t\geq 0$,%
\begin{equation*}
\mathcal{E}(t)\leq c_{1}e^{-c_{2}t},
\end{equation*}%
where $c_{1}$ and $c_{2}$ are positive constants.
\end{theorem}

\begin{proof}
We differentiate (\ref{F(t)m.}), and recall (\ref{ET''}), (\ref{I1'}), and (%
\ref{f4m.}), we obtain%
\begin{eqnarray*}
\mathcal{L}^{\prime }(t) &\leq &\left\{ N\left( \left\vert \mu \right\vert
-2\delta (\xi _{6})/l\right) +\frac{N_{1}}{2\varepsilon _{2}}+m(\xi
_{2})+N_{2}\right\} \int_{0}^{L}u_{t}^{2}dx \\
&&+\left\{ \left( \mu ^{2}l-p(\xi _{1})\right) +\left( \eta +d\right)
\epsilon _{1}\right\} \int_{0}^{L}u_{x}^{2}dx \\
&&+\left\{ -Nk_{2}+N_{1}\left( k_{1}\tau +2\varepsilon _{2}c^{\prime
}+c^{\prime }\right) +\frac{d}{4\epsilon _{1}}\right\} \int_{0}^{L}w^{2}dx \\
&&+\left\{ N_{1}\left( -k_{1}c+3\varepsilon _{2}\right) +\frac{\eta }{%
4\epsilon _{1}}\right\} \int_{0}^{L}\theta ^{2}dx+\left\{ -Nk+\frac{N_{1}}{%
4\varepsilon _{2}}\right\} \int_{0}^{L}\theta _{x}^{2}dx \\
&&+\left\{ -Nk_{3}+\frac{N_{1}}{4\varepsilon _{2}}\right\}
\int_{0}^{L}w_{x}^{2}dx+\left\{ \frac{1}{4}-c^{\prime }N_{2}\right\}
\int_{0}^{L}z^{2}(x,1,t)dx
\end{eqnarray*}

At this point, we choose $\epsilon _{2}$ small enough such that%
\begin{equation*}
-k_{1}c+3\varepsilon _{2}\,<0,
\end{equation*}

then, we choose $\left\vert \mu \right\vert $ small enough such that%
\begin{equation*}
\left\vert \mu \right\vert -2\delta (\xi _{6})/l<0,\text{ }\mu ^{2}l-p(\xi
_{1})<0
\end{equation*}

Now, we take $\epsilon _{1}$ small enough such that%
\begin{equation*}
-p(\xi _{1})+\mu ^{2}l+\left( \eta +d\right) \epsilon _{1}<0
\end{equation*}

then, choosing $N_{1}$ large enough so that%
\begin{equation*}
N_{1}\left( -k_{1}c+3\varepsilon _{2}\right) +\frac{\eta }{4\epsilon _{1}}<0.
\end{equation*}

Once $N_{1}$ is fixed, we then choose $N_{2}$ large enough so that%
\begin{equation*}
\frac{1}{4}-c^{\prime }N_{2}<0.
\end{equation*}%
Finally, we then choose $N$ large enough so that
\begin{equation*}
\begin{array}{l}
-Nc^{\prime }+\frac{N_{1}}{2\varepsilon _{2}}+m(\xi _{2})+N_{2}<0, \\
-Nk_{2}+N_{1}\left( k_{1}\tau +2\varepsilon _{2}c^{\prime }+c^{\prime
}\right) +\frac{d}{4\epsilon _{1}}<0, \\
-Nk+\frac{N_{1}}{4\varepsilon _{2}}<0,\text{ }-Nk_{3}+\frac{N_{1}}{%
4\varepsilon _{2}}<0.%
\end{array}%
\end{equation*}

Thus, using (\ref{ett.}), we arrive at%
\begin{equation}
\mathcal{L}^{\prime }(t)\leq -c\mathcal{E}(t),\text{ \ \ }\forall t>0.
\label{53.}
\end{equation}

A combination of (\ref{equm.}) and (\ref{53.}) gives%
\begin{equation}
\mathcal{L}^{\prime }(t)\leq -c_{2}\mathcal{L}(t),\text{ \ \ }\forall t>0.
\label{65.}
\end{equation}

where $c_{2}=c/c_{2}^{\prime },$ a simple integration of (\ref{65.}) over $%
(0,t)$ yields%
\begin{equation*}
c_{1}^{\prime }\mathcal{E}(t)\leq \mathcal{L}(t)\leq \mathcal{L}%
(0)e^{-c_{2}t}\text{ ,\ \ }\forall t>0.
\end{equation*}

Taking $c_{1}=\mathcal{L}(0)/c_{1}^{\prime }$ which completes the proof.
\end{proof}



\end{document}
